Author’s response to reviews

Title: Nutrition in RMDs: is it really food for thought? Focus on rheumatoid arthritis

Authors:
Alessia Alunno (alessia.alunno82@gmail.com)
Elena Nikiphorou (enikiphorou@gmail.com)
Elena Philippou (philippou.e@unic.ac.cy)
Claire Dairen (cidaien@gmail.com)
Dieter Wiek (dieter_wiek@web.de)
Marios Kouloumas (kouloumas.m@cytanet.com.cy)
Maurizio Cutolo (mcutolo@unige.it)

Version: 2 Date: 05 Jan 2020

Author’s response to reviews:

Executive Editor Comments:

Upon consultation with further editorial board members who requested additional revisions to expand on topic of nutrition in RMDs, the Editor is happy to proceed with this debate piece as it is. The work should further highlight the importance of nutrition to both patients and clinicians.

Authors' reply: We thank the Executive editor for the positive comments on our article

Assistant Editor Comments:

1. Abstract
Your abstract should be structured with a background, main body of abstract and conclusions, and should be no longer than 350 words. Please also include a keywords section, which should three to ten keywords.
Background: the context and purpose of the study
Main Body: brief summary of the debate
Conclusions: brief summary and potential implications
Please ensure you apply this format to the abstract in the main text and the submission system also.
Authors' reply: We structured the abstract in the main text and submission system as requested and added a keywords section.

2. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary
files are cited within the text.
Authors' reply: We uploaded our manuscript as a single, final, clean version including also the table.